Author's response to reviews

Title: Efficacy and adverse effects of intravenous lignocaine therapy in fibromyalgia syndrome

Authors:

Jonathan H Raphael (JonRaph@AOL.com)
Jane L Southall (julie.burrows@dudleygoh-tr.wmids.nhs.uk)
Gareth J Treharne (GJ884@psg-fs4.bham.ac.uk)
George D Kitas (g.d.kitas@bham.ac.uk)

Version: 3 Date: 3 Sep 2002

Reviewer makes 2 requests which have been met:
to reinsert the 2 references that were removed from version1 and to add 2 further subheadings to
methods section.

I have been through editors recommendations and abided by these:
ensure author and affiliation linked by superscript;
ensure abstract conforms to requirements;
change table 1 to landscape view;
change figures to a different format( powerpoint) and crop appropriately.